Author’s response to reviews

Title: Choroidal neovascularization emerged right from the focal choroidal excavation in eyes with central serous chorioretinopathy post half-dose photodynamic therapy: A case report

Authors:

Yang Liu (hefei_103@126.com)
Xin Wang (443946719@qq.com)
Min Zhu (Der_winter@163.com)
Gezhi Xu (16111260005@fudan.edu.cn)
Lei Li (drlilei3028@163.com)

Version: 4 Date: 20 Feb 2019

Author’s response to reviews:

Dear editors and reviewers,

Thanks a great deal to you all for raising valuable comments to my work from busy schedule. Your specialized comments widen and deepen my awareness and understanding of the whole structure and contents of my article. We have addressed the comments raised, and the amendments are as followed.

Michael Paul Blair (Reviewer 1):
“The manuscript describes 2 cases of CNV after PDT for CSCR in the setting of focal choroidal excavation. Given that the N is 2 no strong conclusions can be drawn, but it is of interest and treaters should be aware to consider this phenomenon.”

I can not agree with the reviewer any more that no strong conclusion can be drawn from only two cases. To verify the safety profile in such cases, a prospective, case-controlled clinical trial with a larger sample size is needed. And I remove the part of subjective in the conclusion, only report a special phenomenon in this article.

A revised version has been re-submitted for your consideration. We hope that the revision is acceptable, and are also willing to response to your new comments. Looking forward to hearing from you soon.

Yours sincerely,

Lei Li